Author's response to reviews

Title: Improvement in health-related quality of life in osteoporosis patients treated with teriperatide

Authors:

Arthur N Lau (lauarthur@yahoo.com)
Sammy H Ali (sammy.ali@learnlink.mcmaster.ca)
Anna M Sawka (amsawka@yahoo.com)
Lehana Thabane (thabanl@mcmaster.ca)
Alexandra Papaioannou (PAPAIOANNOU@HHSC.CA)
Amiram Gafni (gafni@mcmaster.ca)
Jonathan D Adachi (jd.adachi@sympatico.ca)

Version: 6 Date: 31 October 2008

Author's response to reviews:

All of the editor's recommendations for formatting changes have been made. Also, please change Dr. JD Adachi's contact information to Ontario (under province, instead of Alberta).